Author's response to reviews

**Title:** Arterial Hypertension due to giant primary adrenal hydatid cyst: Report of a case and review of the literature.

**Authors:**

Tazi M Fadl (tazifadl@yahoo.fr)
Mustapha Ahsaini (drahsaini@gmail.com)
Khallouk Abdellhak (khallabdel@yahoo.fr)
Mellas Soufiane (mellassoufiane@yahoo.fr)
Elfassi M Jamal (jamalfassi@yahoo.com)
Farih M Hassan (prfarih@yahoo.fr)

**Version:** 2 **Date:** 2 September 2011

**Author's response to reviews:**

This case is very interesting, firstly it’s rare location of hydatid cyst, secondly the association with arterial hypertension is very rare also, and third the size of the cyst was never reported in the literature.

My manuscript was previously submitted to BMC Infectious Diseases, and they advise me to contact you for submitting my paper in your journal.

I’m a medical student, and I’m from a country which a monthly income is very low, and I don’t have the money requested in your institution.